Request for Account on Relativity Group SGI Machines

Please read the notes below before filling out the form, and please print VERY clearly. Full Name:

UBC Phone Number (if any):

Preferred account name (8 alphabetic characters maximum): Alternate account name (8 alphabetic characters maximum): Initial Password (please see notes below): E-mail address:

Notes:

ffl Please choose your initial password in accord with the following:

1. Use precisely 8 characters 2. Use at least two non-alphanumeric characters (!, @, #, $, etc.)

3. Do not use the password you intend to use “permanently”; the password you give here should be changed immediately when you first log on to one of the Center machines.

ffl You will be notified in class when your accounts are ready for use.